Author's response to reviews

Title: Web-based smoking cessation intervention that transitions from inpatient to outpatient: study protocol for a randomized controlled trial

Authors:

Kathleen F Harrington (kharring@uab.edu)
Julie A McDougal (julie.mcdougal@ccc.uab.edu)
Maria Pisu (mpisu@uab.edu)
Bin Zhang (binzhang@uab.edu)
Rajani Sadasivam (rajani.sadasivam@umassmed.edu)
Thomas K Houston (Thomas.Houston@umassmed.edu)
William C Bailey (wcbailey@uab.edu)

Version: 4 Date: 16 December 2011

Author's response to reviews: see over
December 15, 2011

Doug Altman, University of Oxford  
Curt Furberg, Wake Forest University of Medicine  
Jeremy Grimshaw, Ottawa Health Research Institute  
Peter Rothwell, University of Oxford  
Editors-in-Chief of Trials

RE: Resubmission of CHART Collaborative Manuscript: Web-based smoking cessation intervention that transitions from inpatient to outpatient: study protocol for a randomized controlled trial

Dear Sirs,

As the lead author on the above-referenced manuscript, I thank your accepting this manuscript for publication. I have reviewed the instructions to authors for Trials and have made appropriate changes to reflect these. As well, we have revised the title as suggested.

Please contact me if anything additional is needed or there are any questions.

Sincerely,

Kathleen F. Harrington, PHD, MPH  
Corresponding Author